Author's response to reviews

Title: Feasibility of brief psychological distress screening by a community-based telephone helpline for cancer patients and carers.

Authors:

Anna L Hawkes (annahawkes@cancerqld.org.au)
Karen L Hughes (k.hughes5@uq.edu.au)
Sandy D Hutchison (sandyhutchison@cancerqld.org.au)
Suzanne K Chambers (SuzanneChambers@cancerqld.org.au)

Version: 7 Date: 8 January 2010

Author's response to reviews: see over
Dear Editor,

Manuscript ID: 3598847282899113 - ‘Feasibility of brief psychological distress screening by a community-based telephone helpline for cancer patients and carers’.

Thank you very much for agreeing to publish the above article and sending through the link to the manuscript proofs. During the review process, the name of the community-based telephone helpline under investigation has changed from ‘Cancer Helpline’ to ‘Cancer Council Helpline’. As such, the authors have revised the document to reflect this change and trust that you will accept these changes that have been tracked in.

Thanks very much for your understanding.

Kind regards

[Signature]

A/Prof Anna Hawkes, PhD
Senior Research Fellow
Viertel Centre for Research in Cancer Control
Cancer Council Queensland